

Citation: Kamau WW, Sang R, Ogola EO, Rotich G, Getugi C, Agha SB, et al. (2022) Survival rate, blood feeding habits and sibling species composition of *Aedes simpsoni complex*. Implications for arbovirus transmission risk in East Africa. PLoS Negl Trop Dis 16(1): e0010171. https://doi.org/10.1371/journal.pntd.0010171

Editor: Felix Hol, Institut Pasteur, FRANCE

Received: October 2, 2021

Accepted: January 14, 2022

Published: January 24, 2022

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pntd.0010171

Copyright: © 2022 Kamau et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Sequences generated in this work have been deposited in the GenBank under accession numbers OK339453 – OK339472 and OK339480 – OK339521 (*Aedes bromeliae*); RESEARCH ARTICLE

Survival rate, blood feeding habits and sibling species composition of *Aedes simpsoni complex*: Implications for arbovirus transmission risk in East Africa

Winnie W. Kamau^{1,2}, Rosemary Sang^{1,3}, Edwin O. Ogola¹, Gilbert Rotich¹, Caroline Getugi¹, Sheila B. Agha¹, Nelson Menza², Baldwyn Torto¹, David P. Tchouassi^{1*}

1 International Centre of Insect Physiology and Ecology, Nairobi, Kenya, 2 Kenyatta University, Nairobi, Kenya, 3 Center for Virus Research, Kenya Medical Research Institute, Nairobi, Kenya

* dtchouassi@icipe.org

Abstract

Aedes simpsoni complex has a wide distribution in Africa and comprises at least three described sub-species including the vellow fever virus (YFV) vector Ae. bromeliae. To date, the distribution and relative contributions of the sub-species and/or subpopulations including bionomic characteristics in relation to YF transmission dynamics remain poorly studied. In this study conducted in two areas with divergent ecosystems: peri-urban (coastal Rabai) and rural (Rift Valley Kerio Valley) in Kenya, survival rate was estimated by parity in Ae. simpsoni s.l. mosquitoes sampled using CO2-baited BG Sentinel traps. We then applied PCR targeting the nuclear internal transcribed spacer 2 (ITS2), region followed by sequencing and phylogenetic analytics to identify the sibling species in the Ae. simpsoni complex among parous and blood fed cohorts. Our results show that Ae. bromeliae was the most dominant sub-species in both areas, exhibiting high survival rates, human blood-feeding, and potentially, high vectorial capacity for pathogen transmission. We document for the first time the presence of Ae. Iilii in Kenya and potentially yet-to-be described species in the complex displaying human feeding tendencies. We also infer a wide host feeding range on rodents, reptile, and domestic livestock besides humans especially for Ae. bromeliae. This feeding trend could likely expose humans to various zoonotic pathogens. Taken together, we highlight the utility of genotype-based analyses to generate precision surveillance data of vector populations for enhanced disease risk prediction and to guide cost-effective interventions (e.g. YF vaccinations).

Author summary

Yellow fever (YF) remains a significant public health risk in East Africa, however, with gaps in the transmission ecology. Important YF virus vectors include *Aedes simpsoni* mosquitoes that comprise subspecies with varying vectoring abilities and, poorly described

OK339522 – OK339529 (Aedes lilii); OK339473 – OK339479 (Aedes simpsoni s.l.).

Funding: WWK received an ICIPE MSc-**Dissertation Research Internship Programme** (DRIP) scholarship through the project Combatting Arthropod Pests for better Health, Food and Climate Resilience (CAP-Africa; project number RAF-3058 KEN-18/0005) funded by Norwegian Agency for Development Cooperation (Norad). Financial support for the study came from the Norad funded CAP-Africa project. DPT is also supported by a Wellcome Trust International Intermediate Fellowship (grant number 222005/Z/ 20/Z). We also gratefully acknowledge the financial support and technical support of ICIPE core donors: Swiss Agency for Development and Cooperation (SDC), Switzerland; Swedish International Development Cooperation Agency (Sida), Sweden; Ministry of Higher Education, Science and Technology, Kenya; and Government of the Federal Democratic Republic of Ethiopia. The views expressed herein do not necessarily reflect the official opinion of the donors. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

ecology and biologic traits relevant to disease transmission. Through active surveillance, we analyzed the survival, human blood feeding habits and genetics of wild populations of *Aedes simpsoni s.l.* in two contrasting ecosystems in Kenya: peri-urban, coastal Rabai, and rural, Rift Valley Kerio Valley. Our findings reveal i) *Aedes bromeliae* as the most abundant subspecies in both areas exhibiting high survival rates, human blood-feeding, and potentially, high vectoring ability, ii) occurrence of *Ae. lilii* contrary to previous reports albeit in low numbers, iii) potential undescribed species in the group displaying human feeding tendencies. Knowledge of the locally adapted subspecies and associated traits that underlie vectorial capacity, impinges on YF distribution risk useful for guiding vector control or cost-effective vaccination.

Introduction

Yellow fever (YF) is a re-emerging arboviral threat in Africa and South America despite availability of an effective vaccine to protect humans. This is unparalleled in eastern Africa exemplified by increased frequency and magnitude of outbreaks recorded recently in Sudan (2012), South Sudan (2020), Uganda (2011, 2016, 2019, 2020), and Ethiopia (2012–2014, 2018; 2020) [1,2]. Imported cases into Kenya following recent outbreaks in Angola and the Democratic Republic of Congo (DRC) (2015–2016) [2] demonstrate the potential for continued YF spread in the region. Improved understanding of the drivers of YF virus (YFV) transmission are urgently required through risk assessment to inform cost-effective preventive strategies.

Of the recognized YFV transmission cycles (sylvatic, rural, and urban cycles), outbreaks in the eastern African region have been described as sylvatic [3]. This is inclusive of the last documented YF outbreak in Kenya (1992–95) that implicated sylvatic *Aedes* vectors including *Aedes simpsoni s.l.* [4]. However, YFV transmission trends appear to be changing with demographic and environmental changes. For instance, the 2015/16 outbreaks in Angola and DRC were urban [2] and whether the East Africa region may face same fate in the future remains uncertain. The abundance of *Ae. simpsoni s.l.* vectors in peri-domestic human environments and associated Stegomyia indices have raised concerns about potential urban/rural risk of YFV transmission in Kenya [5]. *Aedes simpsoni s.l.* thus, has the potential to serve as bridge vector, moving the YFV from the sylvatic/rural to the urban transmission cycle. *Aedes simpsoni* complex comprises at least three described species including; *Ae. lilii, Ae. simpsoni s.s.*, and the YF vector *Ae. bromeliae* [6]. To date, the distribution and relative contributions of the sub-species and/or subpopulations including bionomic characteristics in relation to YFV transmission dynamics remain poorly studied.

Demographic and societal changes including unplanned urbanization as occurring in Africa [7], are among important drivers of vector-borne disease spread. The consequent changes in urban architecture, via effects on environment, human abundance, vector density, and biting behavior have the potential to alter the vectorial capacity of mosquitoes and diseases transmission risk. We here, report data on survival and host feeding patterns of *Ae. simpsoni* mosquitoes in two areas of contrasting ecosystems: peri-urban, coastal Rabai, and rural, Rift Valley Kerio Valley, in Kenya. These parameters provide opportunities for vector infection and efficient transmission and are the two most sensitive indicators of disease transmission potential (i.e. vectorial capacity) [8,9]. Because, human vector feeding must precede outbreaks, knowledge of vector host-feeding behavior is essential to our understanding of the inter-epidemic maintenance patterns of YF. Furthermore, we assessed how these bionomic attributes vary among subspecies of the *Ae. simpsoni* complex.

Materials and methods

Ethics statement

The study received approval from the Scientific Ethics Review Unit (SERU) of the Kenya Medical Research Institute (Protocol NO. SSC 2787). Additionally, consent was sought verbally from household heads to set up traps around their homesteads.

Study site

Adult female *Aedes simpsoni* senso lato (*s.l.*) (hereafter as *Ae. simpsoni*) mosquitoes that had been collected from Rabai (peri-urban environment; human density ~600/km²) and from rural/sylvatic Kerio Valley (KV) (human density ~ 45/km²) as part of an arbovirus surveillance project were used in this study. Rabai (Kilifi County) is located northeast and ~25 km from Mombasa City in coastal Kenya while Kerio Valley with a history of YF outbreak [4] in the Rift Valley (Fig 1). The mosquito breeding habitats differ with Kerio Valley providing rural woodland setting with numerous tree holes for mosquito breeding as opposed to Rabai, with the peri-urban setting providing typical water storage containers for mosquitoes to breed and also outdoors breeding in plant axils and water receptables. Kerio Valley is an arid to semi-arid ecology that is sparsely populated. The main economic activities of the inhabitants include farming of crops like maize and cotton, and livestock keeping especially goats and cattle. Rabai is one of the seven administrative sub-counties of Kilifi county. The main economic activities in the area include subsistence agriculture, casual labor, crafts and petty trading.

Mosquito survey

Adult host seeking female mosquitoes were surveyed using CO₂-baited BG Sentinel traps at three trapping periods with trapping exercise beginning about three weeks after the area first received rainfall continuously for at least a week. This timing was critical to maximize assessment of vector abundance, considering the predominant tree hole and plant axils breeding ecology of Ae. simpsoni [10]. The trapping periods were August-September 2019 (mean daily temperature = 25.4° C; mean daily rainfall = 0.58mm; relative humidity = 75.9%) and February 2020 (mean daily temperature = 27.4°C; mean daily rainfall = 0.30mm; relative humidity = 80.3%) in Rabai and November 2019 (mean daily temperature = 19.9°C; mean daily rainfall = 3.76 mm; relative humidity = 72.4%) in KV. This trap type suitably targets *Stegomyia* mosquitoes which are active during the day [11]. Traps were set from morning to evening (06:30-18:00) on the same day, for at least 8 consecutive days per trapping period in peridomestic areas around homesteads. Sampling was designed to cover a large spatial area within each site. Traps were placed outdoors in the vegetation around human habitations and moved every other day to a new locality at least 500 m away. After retrieval, the mosquitoes were immobilised using triethylamine, sorted and transported in liquid nitrogen from the field to the Emerging Infectious Diseases laboratory at ICIPE, Nairobi. The mosquitoes were later identified morphologically to species level using published taxonomic keys [12] before storing in -80°C freezers for later use.

Parity rate, daily survival and longevity estimation

Mosquitoes after retrieval from -80°C were allowed to thaw on ice (4°C) and then dissected for parity after observing the degree of dilation of the tracheolar skeins of the avarioles [13]. After dissection, the remaining portion of each mosquito was preserved and processed for DNA extraction and molecular speciation (described below). Daily survival rates were derived from estimated parity rates for each sampling period as described previously [14] based on the





https://doi.org/10.1371/journal.pntd.0010171.g001

formula: $P^n = M$, where P is the daily survival rate, *M* the parity rate and *n* the gonotrophic cycle (the number of days between emergence of adult females and first oviposition). A value of 3 days was assumed for this species [9]. Finally, the longevity (days) was estimated using the formula: 1/-lnL where L is the estimated survival [15].

Analysis of blood meals

Blood-fed specimens were individually dissected by separating the abdomen containing engorged blood from the head/thorax. The head/thorax were preserved for each mosquito and subjected to DNA extraction and molecular speciation (described below). Genomic DNA was extracted from the abdomen using the ISOLATE II Genomic DNA Kit (Bioline, Meridian Bioscience, Germany) as per the manufacturer's instructions. DNA was amplified targeting a 500 bp fragment of the 12S mitochondrial rRNA gene using the primers 12S3F [5'-GGGATTAGA-TACCCCACTATGC-3'] and 12S5R [5'-TGCTTACCATGTTACGACTT-3'] [16] and as

described previously [9]. PCRs in a 10 µl reaction volume comprised 2 µl 2x MyTaq Mix (Bioline, Meridian Bioscience, Germany), 10 µM of each primer, 0.2 U of Mytaq DNA polymerase and 1 µl of the template DNA (~20 ng). Thermal cycling conditions were 95°C for 3 min followed by 40 cycles at 95°C for 20 s, 59°C for 30 s and 72°C for 30 s and 72°C for 7 min. Amplicons were resolved on 1.2% agarose gel electrophoresis against a 100bp DNA HyperLadder (Bioline, Meridian Bioscience, Tennessee, USA). The PCR products were purified using the SureClean Plus kit (Bioline, Meridian Bioscience) and outsourced to Microgen (Netherlands) for Sanger sequencing using the forward primer. DNA sequences were compared using the BLAST algorithm and the GenBank database (http://blast.ncbi.nlm.nih.gov/ Blast.cgi). Species level identification was determined when sequences exhibited \geq 98% identity spanning at least 300 bp as described previously [9].

Species typing of mosquitoes

Genomic DNA was extracted from the head/thorax of each blood fed specimen and remaining portion of selected parous mosquitoes as described previously. PCR was performed to identify the sibling species of *Ae. simpsoni* targeting the nuclear internal transcribed spacer 2 (ITS2) region using the primers ITS2A (5'-TGTGAACTGCAGGACACAT-3') and ITS2B (5'-TATGCTTAAATTCAGGGGGGT-3') [17]. ITS2 is fast evolving as a marker of choice widely used for species-level discrimination [18–20] with vast sequence representation in public databases (e.g. GenBank). PCRs were performed in 20µl reaction volumes including 3µl 5x HOT FIREPol Blend Master Mix Kit (Solis Biodyne, Tartu, Estonia), 0.5µl of 10 µM forward and reverse primers, of 5x Mytaq HS mix polymerase and 20 ng DNA template in a ProFlex PCR systems thermocycler (Applied Biosystems, Foster City, CA, USA). The thermal cycling conditions were 95°C for 15 min followed by 40 cycles at 95°C for 30 s, 60°C for 30 s and 72°C for 45 s and 72°C for 7 min. Amplicons were confirmed by gel electrophoresis as described previously. Similarly, the PCR products were purified using SureClean Plus kit and outsourced for Sanger sequencing, in both the forward and reverse direction.

Sequences were viewed and edited in Chromas, embedded in MEGA v.6 [21] prior to querying the GenBank using BLASTn (www.ncbi.nlm.nih.gov/blast). Multiple sequence alignments of the resulting contiguous sequences were performed using ClustalW in MEGA v.6 with default parameters. Maximum likelihood (ML) trees were constructed with nodal support for the different groupings evaluated through 1000 bootstrap replications utilizing the Jukes and Cantor as best-fit model of sequence evolution.

Statistical analyses

The parity rate calculated as the percentage of parous mosquitoes to the total number dissected was established for each trapping period and comparisons made by Pearson chi-squared tests. The 95% confidence intervals (CIs) for the parity rates were estimated using *binom.confint* function. The human blood index is expressed as the proportion of blood-feeding on humans of the total number of blood-fed mosquitoes examined and established for subspecies and tested for significant differences using Pearson chi-squared tests. All analyses were performed at P = 0.05 using R v. 4.0.4 software [22].

Results

Parity, daily survival and age estimates

Overall parity rate was 85% (458/539; 95% CI 81.7–87.7%) broken down as follows: Rabai: August-September 2019 = 82.6% (71/86; 95% CI 73.2–89.1%); February 2020 = 85.1% (189/

Site	Survey period	Parity (% (n))	Daily survival rate	Longevity (days)
Rabai	August-September 2019	82.6 (86) ^a	0.9387	15.8
	February 2020	85.1 (222) ^a	0.9483	18.8
Kerio Valley	November 2019	85.7 (231) ^a	0.9504	19.7
· · ·				

Table 1. Estimated parity, daily survival and age for Ae. simpsoni mosquitoes.

a, not significant

https://doi.org/10.1371/journal.pntd.0010171.t001

222; 95% CI 79.9–89.2%); KV: November 2019 = 85.7% (198/231; 95% CI 80.6–89.6%). The parity rate did not differ between the sites or sampling periods (P>0.05; Table 1). Estimated daily survival rates for *Ae. simpsoni* were high across the sampling periods in Rabai translating to longevity ranging from 15.8–18.8 days (Table 1). *Aedes simpsoni* in Kerio Valley had a higher longevity compared to Rabai, though the difference was not significant (P>0.05; Table 1).

Genetic composition of parous Ae. simpsoni s.l.

A subset of parous specimens (n = 43; Rabai = 37; KV = 6) randomly selected were profiled for DNA sequencing and phylogenetic analysis of the ITS2 region. Our findings show that most of the samples clustered with reference sequences of *Ae. bromeliae* with a strong bootstrap support (73%) (Fig 2). One specimen from Rabai grouped together with *Ae. lilii* in the Genbank and *Ae. simpsoni* samples included from Kajulu, in Kisumu County, western Kenya (Fig 2).

Blood meal feeding patterns

A total of 27 blood fed *Ae. simpsoni s.l.* mosquitoes were analysed from Kerio Valley (n = 9) and Rabai (n = 18), of which blood meal sources were successfully identified from 24 specimens. These represented 9 specimens from Kerio Valley and 15 from Rabai. The data revealed a total of 10 different hosts belonging to fairly large mammals (humans, goat, cow, domestic cat), rodents (grass rat, squirrel, mastomys mouse, African giant pouched mouse, mongoose) and reptile (lizard) (Fig 3). The host range was more diverse in Rabai than Kerio Valley (Fig 3). In both areas, *Ae. simpsoni* had fed more on humans, then squirrel (Rabai) followed by a lower representation of other host types (Fig 3). The overall human-blood-index (HBI) was 0.33 (8/24).

Association of Ae. simpsoni subspecies and human blood feeding

Next, we examined the relationship between subspecies of *Ae. simpsoni s.l.* and their influence on human blood feeding. A total of 27 *Ae. simpsoni sl* blood-fed specimens were profiled for phylogenetic analysis of sequenced ITS-2 region, with most of samples obtained from Rabai (n = 18), then Kerio Valley (n = 9). ITS2 sequences of the three blood-fed samples with unsuccessful blood meal data were included. The *Ae. simpsoni sl* samples resolved into 3 clades, with well supported bootstrap values (Fig 4). One of these clustered with *Aedes bromeliae* (GenBank No: KF135509) and had most of the samples from Rabai (n = 11) and Kerio Valley (n = 6). Clade II contained samples exclusively from Rabai (n = 4), while clade III had samples solely from Kerio Valley (n = 3). Thus, our findings show overwhelming representation of *Ae. bromeliae* among the blood-fed samples. Each of the clades exhibited human feeding tendency with no variation in the estimated human blood index (HBI) between the clades (P>0.05); clade I which clustered with *Ae. bromeliae*: (4/18); clade II: 2/3; clade III: 2/3 (Fig 4).





https://doi.org/10.1371/journal.pntd.0010171.g002

Discussion

Here, we provide contemporary estimates of survival and blood feeding patterns for the important arboviral vector *Ae. simpsoni s.l.* collected from two areas with distinct ecosystems. High parity rates and hence survival rates/longevity were evident for this mosquito across the





https://doi.org/10.1371/journal.pntd.0010171.g003

three trapping periods. *Aedes simpsoni s.l.* from Kenya was previously reported to be susceptible to YFV [23], and because infectious mosquitoes reflect the age structure of adult female mosquitoes [24], our findings suggest a high vectorial capacity for pathogen transmission by this mosquito species despite moderate human feeding rates.



Fig 4. Maximum-likelihood tree derived from *Aedes simpsoni* **blood-fed ITS2 sequences using a JC model (255–313 nt).** Bootstrap values are shown above relevant nodes. Sequence of *Ae. aegypti* indicated as outgroup. The scale-bar indicates the number of substitutions per site. Taxon abbreviations represent sampling sites with numbers corresponding to specific sequence samples: RAB, Rabai; KV, Kerio Valley. Sequences have been submitted to GenBank with accession numbers OK339453–OK339472 (*Aedes bromeliae*), OK339473-OK339479 (*Aedes simpsoni* sl).

https://doi.org/10.1371/journal.pntd.0010171.g004

Ae. bromeliae, which is the only known YFV vector among *Ae. simpsoni* mosquitoes in East Africa [6,25], has equally been suggested as the only human biting sibling species occurring in Kenya [6]. *Ae. bromeliae* appears dominant in the study areas based on our parity data (ran-domly selected) and blood fed specimens. While our result agrees with data reported in the literature [6,25], this trend might have been affected by its domestic/peri-domestic habits where sampling was conducted. The high survival/longevity found in the current study may contribute to the vectorial capacity of *Ae. bromeliae*. Age is the most sensitive vectorial capacity (a measure of pathogen transmission potential) parameter [8,9]. Higher YFV transmission risk relating to this subspecies is therefore, suggested in Kenya, although validation of its competence is still an important knowledge gap. *Ae. simpsoni* mosquitoes commonly breed in tree holes/plant leaf axils [5]. Its adaptation to breeding in water holding containers around human habitations [5] may contribute to seasonal abundance to enhance human risk as a result of increased vector human contact.

Our sequencing and phylogenetic analyses document for the first time the presence of Ae. *lilii* in Rabai, Kenya, thereby confirming previous speculations on the existence of this subspecies in the area [26]. The presence of Ae. lilii in Kenya was further supported by data from Kajulu, Kisumu in western Kenya (Fig 2). The results suggest geographic differences in the distribution of the sibling species in Kenya, warranting further studies to investigate this hypothesis. The very low numbers and absence of blood feeding for Ae. lilii, suggest minimal importance of this non-human biter [6,25], in disease transmission to humans in the study areas. Notably, we uncover potentially undescribed species (clades II and III) within the complex based on phylogenetic resolutions which displayed human feeding tendencies (Fig 4). The diversity of species in this complex could be much higher than previously thought which would require additional confirmatory studies. These studies could incorporate additional markers including analysis at the genomic level. Additionally, studies on the distribution and detailed ecology that could facilitate assessment of arboviral disease risk associated with distinct subspecies are warranted. Furthermore, a PCR-based method to distinguish the vector Ae. bromeliae from the non-vector Ae. lilii based on the ITS region was recently described [25]. Application of this protocol to our samples gave mixed and inconclusive results because several specimens amplified with both specific primer sets targeting these species.

Blood meal data showed feeding on diverse hosts for *Ae. simpsoni* with a corresponding moderate human blood index, a trend largely driven by *Ae. bromeliae*. The overall engorged specimens analyzed was relatively small (n = 24) reflecting the difficulty to trap blood-fed cohorts. This could as well have been affected by our trapping approach possibly biased towards host-seeking females. However, we attempted the newly developed *Aedes* gravid traps to enhance gravid and engorged mosquito cohorts [27] but had no catches. Thus, we suggest that further studies should incorporate resting collections to enhance the number of freshly engorged mosquitoes [28]. While a large sample size could allow for better inference on the trophic habit of these species, nonetheless, our study provides useful baseline data regarding host utilization by this heterogenous species group inferring opportunistic feeding in these arboviral disease foci. Feeding on humans by *Ae. bromeliae* and the potential uncharacterised species, increases the risk of transmission to humans of diverse vector-borne pathogens (e.g., yellow fever, dengue, chikungunya, and Zika viruses) including zoonotic ones circulating in livestock or rodent hosts.

YF outbreak preparedness and the Eliminate Yellow Fever Epidemics (EYE) strategy largely centers on vaccination scale-up, a costly venture that must be guided by justified risk. This focus must expand to include mosquito-based surveillance of critical parameters that define vectorial capacity in specific ecological contexts [8,9]. Combining laboratory experiments and field ecology while incorporating genotype-based analysis of vector populations should

generate integrated data that could be modeled to predict potential YFV spread and re-emergence risk.

We conclude that blood-fed and parous specimens of *Ae. simpsoni s.l.* mosquitoes in the study areas were mainly *Ae. bromeliae* (the primary YFV vector in East Africa), and it appears to be the most dominant species in this complex. High parity rates and hence survival rates/longevity were evident for this mosquito across the three trapping periods, suggesting a high vectorial capacity for pathogen transmission by this mosquito. We report the presence of *Ae. lilii* in Kenya with potentially yet-to-be described species in the complex (clades II and III, Fig 4) displaying human feeding tendencies. We also infer a wide host feeding range on rodents, reptile, domestic livestock besides humans especially for *Ae. bromeliae*, a feeding trend that could likely expose humans to various zoonotic pathogens. Overall, we highlight the importance of precision surveillance data of vector populations through genotype-based analyses for enhanced disease risk prediction and to guide cost-effective interventions (e.g. YF vaccinations).

Acknowledgments

We thank the chiefs and community members of the study sites for their cooperation and support.

Author Contributions

Conceptualization: Rosemary Sang, David P. Tchouassi.

Data curation: Winnie W. Kamau, Gilbert Rotich, David P. Tchouassi.

Formal analysis: Winnie W. Kamau, Edwin O. Ogola, David P. Tchouassi.

Funding acquisition: Rosemary Sang, Baldwyn Torto, David P. Tchouassi.

Investigation: Winnie W. Kamau, Rosemary Sang, Edwin O. Ogola, Gilbert Rotich, Caroline Getugi, Sheila B. Agha, David P. Tchouassi.

Methodology: Winnie W. Kamau, David P. Tchouassi.

Project administration: Rosemary Sang, Baldwyn Torto, David P. Tchouassi.

Resources: Rosemary Sang, Baldwyn Torto, David P. Tchouassi.

Supervision: Rosemary Sang, Nelson Menza, David P. Tchouassi.

Validation: Rosemary Sang, Nelson Menza, Baldwyn Torto, David P. Tchouassi.

Visualization: David P. Tchouassi.

Writing - original draft: Winnie W. Kamau, David P. Tchouassi.

Writing – review & editing: Winnie W. Kamau, Rosemary Sang, Edwin O. Ogola, Gilbert Rotich, Caroline Getugi, Sheila B. Agha, Nelson Menza, Baldwyn Torto, David P. Tchouassi.

References

- 1. WHO Disease Outbreak News (DONs). https://www.who.int/emergencies/disease-outbreak-news/4.
- 2. WHO Regional Office for Africa. Yellow fever. https://www.afro.who.int/health-topics/yellow-fever.
- Ellis BR, Barrett AD. The enigma of yellow fever in East Africa. Rev Med Virol. 2008; 18(5):331–46. https://doi.org/10.1002/rmv.584 PMID: 18615782

- Reiter P, Cordellier R, Ouma JO, Cropp CB, Savage HM, Sanders EJ, et al. First recorded outbreak of Yellow fever in Kenya, 1992–93 II. Entomologic Investigations. Am J Trop Med Hyg. 1998; 59(4): 650– 656. https://doi.org/10.4269/ajtmh.1998.59.650 PMID: 9790447
- Agha SB, Tchouassi DP, Bastos AD, Sang R. Assessment of risk of dengue and yellow fever virus transmission in three major Kenyan cities based on Stegomyia indices. PLoS Negl Trop Dis. 2017; 11 (8):e0005858. https://doi.org/10.1371/journal.pntd.0005858 PMID: 28817563
- Huang YM. Aedes (Stegomyia) bromeliae (Diptera: Culicidae), the yellow fever vector in East Africa. J Med Entomol. 1986; 23:196–200. https://doi.org/10.1093/jmedent/23.2.196 PMID: 3701804
- Wilder-Smith A, Gubler DJ, Weaver SC, Monath TP, Heymann DL, Scott TW. Epidemic arboviral diseases: priorities for research and public health. Lancet Infect Dis. 2017; 17(3):e101–6. https://doi.org/10.1016/S1473-3099(16)30518-7 PMID: 28011234
- Kramer LD, Ebel GD. Dynamics of flavivirus infection in mosquitoes. Adv Virus Res. 2003; 60:187– 232. https://doi.org/10.1016/s0065-3527(03)60006-0 PMID: 14689695
- Tchouassi DP, Torto B, Sang R, Riginos C, Ezenwa VO. Large herbivore loss has complex effects on mosquito ecology and vector-borne disease risk. Transbound Emerg Dis. 2021; 68(4):2503–2513. https://doi.org/10.1111/tbed.13918 PMID: 33170555
- Bown DN, Bang YH. Ecological studies on Aedes simpsoni (Diptera: Culicidae) in southeastern Nigeria. J Med Entomol. 1980; 17(4): 367–374. https://doi.org/10.1093/jmedent/17.4.367 PMID: 7420363
- Agha SB, Tchouassi DP, Bastos AD, Sang R. Dengue and yellow fever virus vectors: seasonal abundance, diversity and resting preferences in three Kenyan cities. Parasit Vectors. 2017; 10(1): 628. https://doi.org/10.1186/s13071-017-2598-2 PMID: 29284522
- 12. Edwards FW. *Mosquitoes of the Ethiopian Region. III.-Culicine Adults and Pupae*, London, UK: British Museum (Nat. Hist.).1941. 499 pp.
- 13. Detinova TS. Age grouping methods in Diptera of medical importance with special reference to some vectors of malaria. Monograph Series World Health Organization. 1962; 47:13–191. PMID: 13885800
- 14. Davidson G. Estimation of the survival-rate of anopheline mosquitoes in nature. *Nature*. 1954; 174 (4434):792–793. https://doi.org/10.1038/174792a0 PMID: 13214009
- Clements AN, Paterson GD. The analysis of mortality and survival rates in wild populations of mosquitoes. J Appl Ecol. 1981; 18:373–399.
- Roca AL, Bar-Gal GK, Eizirik E, Helgen KM, Maria R, Springer MS, et al. Mesozoic origin for West Indian insectivores. Nature. 2004; 429(6992):649–51. <u>https://doi.org/10.1038/nature02597</u> PMID: 15190349
- Beebe NW, Saul A. Discrimination of all members of the Anopheles punctulatus complex by polymerase chain reaction-restriction fragment length polymorphism analysis. Am J Trop Med Hyg. 1995; 53:478– 81. https://doi.org/10.4269/ajtmh.1995.53.478 PMID: 7485705
- Tchouassi DP, Bastos AD, Sole CL, Diallo M, Lutomiah J, Mutisya J, et al. Population genetics of two key mosquito vectors of Rift Valley fever virus reveals new insights into the changing disease outbreak patterns in Kenya. PLoS Negl Trop Dis. 2014; 8:e3364. https://doi.org/10.1371/journal.pntd.0003364 PMID: 25474018
- Beebe NW. DNA barcoding mosquitoes: advice for potential prospectors. Parasitology. 2018; 145 (5):622–633. https://doi.org/10.1017/S0031182018000343 PMID: 29564995
- Ogola EO, Chepkorir E, Sang R, Tchouassi DP. A previously unreported potential malaria vector in a dry ecology of Kenya. Parasit Vectors. 2019; 12:80. https://doi.org/10.1186/s13071-019-3332-z PMID: 30744665
- Tamura K, Stecher G, Peterson D, Filipski A, Kumar S. MEGA6: Molecular Evolutionary Genetics Analysis version 6.0. Mol Biol Evol. 2013; 30(12):2725–2729. https://doi.org/10.1093/molbev/mst197 PMID: 24132122
- R Core Team (2020). R: A language and environment for statistical computing. R Foundation for Statistical Computing, Vienna, Austria. URL https://www.R-project.org/.
- Ellis BR, Sang RC, Horne KM, Higgs S, Wesson DM. Yellow fever virus susceptibility of two mosquito vectors from Kenya, East Africa. Trans R Soc Trop Med Hyg. 2012; 106(6):387–9.24. https://doi.org/ 10.1016/j.trstmh.2012.02.007 PMID: 22521217
- Smith DL, Dushoff J, McKenzie FE, Dobson AP. The risk of a mosquito-borne infection in a heterogeneous environment. PLoS Biol. 2004; 2(11):e368. <u>https://doi.org/10.1371/journal.pbio.0020368</u> PMID: 15510228
- Bennett KL, Linton YM, Shija F, Kaddumukasa M, Djouaka R, Misinzo G, et al. Molecular differentiation of the African yellow fever vector *Aedes bromeliae* (Diptera: Culicidae) from its sympatric non-vector sister species, *Aedes Iilii*. PLoS Negl Trop Dis. 2015; 9(12):e0004250. https://doi.org/10.1371/journal. pntd.0004250 PMID: 26641858

- Walter KS, Brown JE, Powell JR. Microhabitat partitioning of Aedes simpsoni (Diptera: Culicidae). J Med Entomol. 2014; 51(3):596–604. https://doi.org/10.1603/me13097 PMID: 24897852
- Eiras AE, Buhagiar TS, Ritchie SA. Development of the Gravid Aedes Trap for the capture of adult female container–exploiting mosquitoes (Diptera: Culicidae). J Med Entomol. 2014; 51(1):200–209. https://doi.org/10.1603/me13104 PMID: 24605470
- Diouf B, Sene NM, Ndiaye EH, Gaye A, Ngom EH, Gueye A, et al. Resting behavior of blood-fed females and host feeding preferences of *Aedes aegypti* (Diptera: Culicidae) morphological forms in Senegal. J Med Entomol. 2021; XX(X), 2021, 1–7. https://doi.org/10.1093/jme/tjab111 PMID: 34165556